Author's response to reviews

Title: Absence of inferior vena cava in 14-year old boy associated with deep venous thrombosis and positive Mycoplasma pneumoniae serum antibodies- a case report.

Authors:

Boleslaw Kalicki (kalicki@wim.mil.pl)
Monika Sadecka (monika.sdc@gmail.com)
Agata Wawrzyniak (awawrzyniak@wim.mil.pl)
Piotr Kozinski (pkozinski@wim.mil.pl)
Miroslaw Dziekiewicz (dziekiewicz@wim.mil.pl)
Anna Jung (ajung@wim.mil.pl)

Version: 3
Date: 29 December 2014

Author's response to reviews:

Thank you for your interest in our case presentation and enclosed opinions of two reviewers. The authors have carefully read both reviews and implemented changes in the manuscripts. Because the 1st review of our article was highly favorable we do not performed any modifications due to it. According to the 2nd review the authors:

- made the Background section less detailed;
- add the reference [2] that confirms the frequency of AIVC from our manuscript;
- removed the longer description of our case from the background section, leaving a brief statement of what is being reported in the report;
- improved the quality of the Fig.1. and added new descriptions;
- rewritten the Conclusion section, explaining why our case is different from previous literature reports in this subject;
- added reference [6], due to the suggestion;
- made most of minor and discretionary revisions- according to the rule stated in the Instructions for authors, we did not use any abbreviations in the Abstract.

We believe, that we gave a satisfying response to all the concerns. In our opinion, the article has an educational value and make a contribution to clinical practice. We look forward to receiving your comments on our efforts.

Yours sincerely,

Boleslaw Kalicki,

Department of Pediatrics,
Pediatric Nephrology and Allergy.
Military Institute of Medicine,
Warsaw, Poland.
kalicki@wim.mil.pl
On behalf of all authors.